Author's response to reviews

Title: An unusual case of ST-segment elevation myocardial infarction following a late bare-metal stent fracture in a native coronary artery: a case report

Authors:

Giovanni Minardi (giovanni.minardi@libero.it)
Paolo G. Pino (ppino@scamilloforlanini.rm.it)
Marco Stefano Nazzaro (nazzaro@scamilloforlanini.rm.it)
Herribert Pavaci (herrpav@hotmail.com)
Martina Sordi (martinasordi@yahoo.it)
Cesare Greco (cesare.greco@uniroma1.it)
Carlo Gaudio (carlo.gaudio@tiscali.it)

Version: 4 Date: 19 November 2009

Author's response to reviews: see over
November 19th, 2009

Michael Kidd, MD
Editor-in-Chief of Journal of Medical Case Reports

Dear Dr. Kidd,

We are submitting you the new version of our manuscript entitled “An unusual case of ST-segment elevation myocardial infarction following a late bare-metal stent fracture in a native coronary artery: a case report” with the requested changes.

Thank you again for the positive consideration,
Yours Sincerely,
Giovanni Minardi